Author’s response to reviews

Title: Characteristics of febrile urinary tract infections in older male adults

Authors:
Alex Smithson (asa30412@hotmail.com)
Javier Ramos (xramos@fhes.cat)
Esther Niño (enino@fhes.cat)
Alex Culla (aculla@fhes.cat)
Ubaldo Perttierra (upertier@fhes.cat)
Michele Friscia (mfriscia@fhes.cat)
Maria Teresa Bastida (mtbastida@fhes.cat)

Version: 3 Date: 15 Nov 2019

Author’s response to reviews:

Barcelona, November 14th, 2019

Rebecca Strong
BMC Geriatrics

Re: Characteristics of febrile urinary tract infections in older male adults (BGTC-D-19-00294R2)

In this new version of the manuscript we have clarified the competing interests as one of the authors is member of the board of BMC Geriatrics. We have also modified the text to minimize the overlap with our previous article. Finally, we have had the text reviewed and improved by a native speaker. We hope that this resubmission addresses all the issues and makes the manuscript suitable for publication in BMC Geriatrics.

Sincerely,

Alex Smithson, MD, PhD
Infectious Diseases Unit
Avinguda Mossen Pons i Rabadà s/n
08923, Santa Coloma Gramenet, Barcelona